Author's response to reviews

Title: Biepicondylar fracture presenting with dislocation in elbow: a case report

Authors:

Savas Guner (gunersavas@gmail.com)
Sukriye Ilkay Guner (ilkay.guner@hotmail.com)
Mehmet Fethi Ceylan (mfethiceylan@hotmail.com)
Gokay Görmeli (ggormeli@yahoo.com)
Cemile Ayse Gormeli (ayseyazici@yahoo.com)
Haci Onder (dr_bethesda@hotmail.com)

Version: 7 Date: 17 April 2012

Author's response to reviews: see over
Dear Editor,

We would like to submit the attached manuscript “**Biepicondylar fracture presenting with dislocation in elbow: a case report**” for your kind consideration. Biepicondylar fracture is *unusual presentation* of elbow fracture. There are only three cases in the English literature. So we believe this report providing data to the literature.

This manuscript has not been submitted to any other journal. We further declare that we have no financial arrangement with any product or rival product featured in the article, sources of funding, institutional affiliations, and any possible financial or personal conflicts of interest. Consent form was taken from her mother with signed and the form was uploaded (in additional material files).

We guarantee that the manuscript will not be published elsewhere in any language without the consent of the copyright owners, that the rights of third parties will not be violated, and that the publisher will not be held legally responsible should there be any claims for compensation.

All co-authors have seen and approved the final version of the paper and accept responsibility for the data presented. The interests of each author for the publication have no differences of opinion.

SG was the main author and performed the clinical assessment, performed the surgery and follow-up. HO performed the clinical assessment and the follow-up. GG performed the bibliographic research. MFC performed the clinical assessment and the surgery. SIG was a major contributor in writing the manuscript and performed cast care. CAG performed the radiological assessment. All authors have read and approved the final manuscript.

Thanking you

Sincerely

**Corresponding author**
Savas Guner MD
Assistant Professor, Yüzüncü Yıl University
School of Medicine
Department of Orthopedy and Traumatology,
65200, Van, Turkey.
Telephone: +90 (533) 243 38 73

**E mail:** gunersavas@gmail.com